



# Article The Cryptocurrency Market in Transition before and after COVID-19: An Opportunity for Investors?

An Pham Ngoc Nguyen <sup>1,2,\*</sup>, Tai Tan Mai <sup>1,3</sup>, Marija Bezbradica <sup>1,3</sup>, and Martin Crane <sup>1,3</sup>

- <sup>1</sup> School of Computing, Dublin City University, Collins Ave Ext, Whitehall, D09 Y074 Dublin, Ireland
- <sup>2</sup> SFI Centre for Research Training in Artificial Intelligence, D02 FX65 Dublin, Ireland
- <sup>3</sup> ADAPT Center for Digital Content Technology, D02 PN40 Dublin, Ireland

Correspondence: ngocan.nguyenpham6@mail.dcu.ie

Abstract: We analyze the correlation between different assets in the cryptocurrency market throughout different phases, specifically bearish and bullish periods. Taking advantage of a fine-grained dataset comprising 34 historical cryptocurrency price time series collected tick-by-tick on the *HitBTC* exchange, we observe the changes in interactions among these cryptocurrencies from two aspects: time and level of granularity. Moreover, the investment decisions of investors during turbulent times caused by the COVID-19 pandemic are assessed by looking at the cryptocurrency community structure using various community detection algorithms. We found that finer-grain time series describes clearer the correlations between cryptocurrencies. Notably, a noise and trend removal scheme is applied to the original correlations thanks to the theory of random matrices and the concept of Market Component, which has never been considered in existing studies in quantitative finance. To this end, we recognized that investment decisions of cryptocurrency traders vary between bearish and bullish markets. The results of our work can help scholars, especially investors, better understand the operation of the cryptocurrency market, thereby building up an appropriate investment strategy suitable to the prevailing certain economic situation.

**Keywords:** cryptocurrencies; noise and trend effects; tick-by-tick data; network structure; community detection; COVID-19

# 1. Introduction

The cryptocurrency market has become an attractive target for many financial investors in recent years due to its potential for rapid gains. One research topic being explored in this market is the correlation between different cryptocurrencies. Understanding how different assets interact with each other can help in portfolio optimization [1], predicting the future volatility or downturn [2] and also in observing the risk spillover that benefits portfolio diversification [3], to mention only a few.

Thanks to a network-based methodology, cryptocurrencies' cross-relationships can be learned and observed visually [4]. The idea of this method is that it builds up a network of different objects such that the distance between two objects depends on how similar they are: the shorter the distance, the more similar the two objects are. Eventually, we can see the interaction between objects by looking at their network's structure and analyzing characteristics of the network. Different network construction approaches have been explored in the literature, from Minimum Spanning Tree (MST) [5], k-Nearest neighbors (kNN) [6], planar maximally filtered graph (PMFG) [2] to Threshold Weighted-Minimum Dominating Set (TW-MDS) [7], to name but a few. In financial markets, normally, the similarity between two assets is measured by comparing the evolution of two corresponding price time series, one typical method to do this is Pearson correlation metric [8]. The study on correlation of traditional asset classes such as stocks, bonds, national fiat currencies and commodities has been developed a long time ago, with varying approaches invented to learn the correlation



Citation: Nguyen, A.P.N.; Mai, T.T.; Bezbradica, M.; Crane, M. The Cryptocurrency Market in Transition before and after COVID-19: An Opportunity for Investors? *Entropy* 2022, 24, 1317. https://doi.org/ 10.3390/e24091317

Academic Editors: Stanisław Drożdż, Jarosław Kwapień and Marcin Watorek

Received: 19 August 2022 Accepted: 14 September 2022 Published: 19 September 2022

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). between different entities in the same market but also between different asset classes, ranging from statistical [9,10] to AI-based methods [11].

Generally, there are two common shortcomings with correlation-related studies. Firstly, one mainly uses a low-frequency dataset such as daily or monthly, and this might cause a loss of important information from each time series, hence failing to reflect their true nature [12]. This appears to be a major concern in the cryptocurrency market, since it is well-known for its high fluctuations in terms of price movement. For example, in [13], the authors show that the losses of cryptocurrencies can reach 70% within one day. Recently, in 2020, by comparing the volatility in the returns between cryptocurrency and stock markets, the authors of [14] revealed that major cryptocurrencies such as BTC and ETH have volatilities of 5.68 and 7.10, respectively, which is two-fold higher than that of S&P500 and Euro Stoxx 50 indices. Notably, Dirk et al. calculated the daily price volatility of Bitcoin from 2001 until 2021 and found that there are extremely volatile days when the volatility can hit 120% [15]. Thus, using a high frequency means that we are ignoring valuable information (e.g., the intraday fluctuations of a time series) on purpose. As a result, this can adversely affect the correlation extracted from the dataset, potentially leading to inaccurate correlation-using experiments (e.g., portfolio optimization). Secondly, researchers tend to analyze the inter-relation between different time series by using trading price values reported on a website (e.g., Coinmarket (https://coinmarketcap.com/), Yahoo Finance (https://finance.yahoo.com/)). However, this practice deliberately ignores the effects of noise and trends in financial time series, which we will describe clearly in Section 4.

Another important factor to consider is the recent *COVID-19* pandemic which forced all countries to close off borders and restrict movements for residents as well as businesses [16]. This had a strong effect on the global downturn which occurred in March 2020 as a response to governments' efforts to control the disease spreading [17]. These historical events have been shown to disturb and devalue different financial asset classes such as stocks, bonds and also cryptocurrencies [18,19]. Instead of looking at the changes in time-series elements such as volumes, prices, returns and volatilities during the COVID-19 pandemic, in this study, we will investigate the impact of the pandemic by looking at the changes in *network structures* over time. Furthermore, based on these network's structures, we show how we can observe the corresponding community structures via community detection methods. The results from our experiment can be used to learn behaviours of investors in different periods of time, especially during downturn times in the financial market.

From the shortcomings of existing studies and utilizing the advantage of networkbased analysis, this study aims to investigate the network structure of cryptocurrencies without noise and trend effects and how this structure changes under the impact of the COVID-19 pandemic. Specifically, the research target is to answer these research questions:

- RQ1. Is there evidence of the existence of noise and trend effects in the cryptocurrency market? If yes, how do noise and trend effects influence the interactions between cryptocurrencies? What does the network structure of these cryptocurrencies look like after removing noise and trend effects?
- RQ2. Does the network structure change when the level of granularity changes? If this is the case, what level of granularity should we use to obtain the true network structure?
- RQ3. Is there evidence that historical events such as the COVID-19 pandemic and the global downturn in 2020 changed the overall cryptocurrency network structure? If this is the case, how did they change it? Moreover, is there any possibility that this change was caused by a change in investors' investment strategy? In other words, does the way investors react to a downturn change the interactions between cryptocurrencies?

It should be noted that we are not new to the subject of time-varying cryptocurrency network structure, we merely build on work by the team of Drozdz, Watorek, Kwapien [20,21] as well as, more recently, Nie [22]. However, our work expands the existing studies since we consider the investment decisions of investors based on the observed network structure and we acknowledge the negative effect of not only trend but also noise presenting in cryptocurrencies. As suggested by Miceli [23], the trend and noise removal results in a filtered MST that better explains investment strategy and also potentially uncovers endogenous or exogenous factors that drive the price of cryptocurrencies

To solve these research questions, we use a tick-by-tick dataset which consists of 34 price time series corresponding to 34 cryptocurrencies traded on the HitBTC exchange during the period between 13 February 2019 and 6 April 2021. When it comes to network formation, we calculate the correlation between cryptocurrencies by adopting the linear similarity measurement named Pearson and then construct a Minimum Spanning Tree (MST) based on these correlation coefficients. The noise and trend removal is carried out by applying Random Matrix Theory (RMT). Community structure is found by using community detection methods. In addition, different metrics are used to analyze the network structures and support our findings.

The remainder of the article is organized as follows: Section 2 presents an overview of the relevant literature. Section 3 provides a description of the dataset. Section 4 describes terminologies, methods and preprocessing procedures. Section 5 discusses the experimental results followed by implications and hypotheses. Finally, the conclusion of this study is given in Section 7.

#### 2. Related Works

#### 2.1. Correlation-Based Analysis in the Financial Markets

The topic of correlation analysis has a long history in connection with stock markets throughout various historical economic crises using different correlation-calculating metrics. In [24], the authors estimated the correlation between 116 S&P500 stocks between 1982 and 2000 using Pearson coefficient. They further used MST to build up a correlation-based network in order to observe time-varying correlations based on three network measuring metrics including normalized tree length, survival ratio and mean occupation layer. As a result, they pointed out a large change in the network structure during Black Monday. More recently, [6] came up with a Neural Network approach to construct a graph and found a dramatic difference in the network structure during the downturns in 2008, 2011 and 2020. In [1], a Pearson correlation matrix of 200 and 400 stocks from the CSI 300 and S&P500 index, respectively, was used to find an optimized portfolio following the Markowitz optimization scheme. Instead of using Pearson method, Liu et al's paper used an interesting alternative method Mutual Information to generate a distance metric to take account of non-linear effects in intra-day S&P stock data [25]. Other methods to estimate the correlation coefficients (i.e., Wavelet coherence, Fast Fourier Transform) and construct correlation-based networks (i.e., PMFG, threshold method) were introduced in several studies [2,11,26].

Different existing approaches to study the correlations in the stock market have been applied to digital coins. Some common conclusions from existing articles are that the cryptocurrency network changes over time but Ethereum tends to act as a central node in the whole network, i.e., it is a densely connected node [5,27,28]. A few works remedy the problem of dataset shortages that have been concerned in the traditional markets, i.e ones tended to use low-frequency data to implement their studies such as daily or weekly. However, they only account for a small portion of the existing literature. For example, Antonio et al. [29] used small frequency resolutions such as one hour and four hours and also consider daily data of 25 large market capitalized entities traded on the FTX exchange to discover the evolution of cryptocurrency network structures between different time frequencies. By using Pearson correlation-based MST, they found an increase in the complexity of networks' shape for coarser time resolutions. In other words, cryptocurrencies converge into a bigger group as resolution increases. On the contrary, the authors in [20] using multiple timescales starting at 10 min to 360 min proposed an opposite statement that low timescales cause the network to be centralized while it is distributed and more correlated at high timescales. They used the liquidity and capitalization differences among the assets to explain this result, since cryptocurrencies

with low capitalization are traded less frequently than those with large capitalization, it takes more time for a piece of market information to spread over such cryptocurrencies. Thus, they are more inclined to use longer scales. Notably, this is one of the very few studies that remove the trend effect from the original dataset. Interestingly, instead of using return time series like other researchers, a research using hourly realized volatility values was carried out to observe the risk spillover between 7 high-capitalized cryptocurrencies [3].

Different methods have been introduced to detect communities given a correlation matrix. The authors in [4] applied Louvain method on the MST of 119 cryptocurrencies to cluster potential communities. The time-varying dynamics from the community structures found suggests collective behaviour among these communities. With the communities found by the same method, the authors in [30] went one step further by using Principal Component Analysis (PCA) to find an optimal portfolio out of 200 cryptocurrencies in circulation. Another community detection method that is worth taking into consideration is Girvan–Newman, which has been adopted widely for multiple purposes such as link prediction, portfolio diversification, etc. [31,32]. A few other methods are also being used to grouping similar entities but are less popular such as Clauset algorithm, Stochastic block model (SBM), Latent Dirichlet Allocation (LDA) and Markov random field (MRF) [33]. One obstacle from existing studies is that some used a specific community detection algorithm only, raising a doubt about the robustness of the community structure. To this end, we first use the Louvain method to detect communities in our dataset and then adopt Girvan–Newman method to examine the robustness of the communities found earlier.

#### 2.2. How the COVID-19 Pandemic Intervened on the Economy Worldwide

At the beginning of 2020, the economy of China started to be influenced by COVID-19, earlier than other countries. Moreover, as the world's hub for global manufacturing and trade, immediate adverse effects on the Chinese economy resulted in global impacts [16]. Different regulations have been applied to handle the disease, such as closing national borders as well as stopping business activities across the world, strongly influencing the global economy [16]. Eventually, the global financial panic in March 2020 took place. In [18], the authors pointed out that the similarity calculated by ACC and ADCC models between the US and Chinese markets increased dramatically during the pandemic. Regarding the stock prices, when the pandemic occurred, the prices of the US and Chinese stocks decreased but started to recover again since July 2020. This trend is also true for other emerging and developed stock markets in different countries from different continents such as Japan, Germany, Australia and Canada [34]. Likewise, even less risky assets such as gold were adversely affected [35]. The increase in the correlation between different financial markets in the presence of good and bad news has been observed for some decades. In [36], the authors stated that stocks are more affected by the presence of bad news, compared to good news. Moreover, bad news has a stronger correlation in traditional markets. These results align with what happened during the COVID-19 pandemic. Although the world continued facing different COVID-19 waves afterwards, its impact on different asset classes lessened significantly [37], stock prices increased and volatilities decreased again to their original values before the pandemic [38]. Furthermore, the connectedness between different assets also experienced a major decline [39].

In [19], the authors investigated the impact of the COVID-19 pandemic on the cryptocurrency market by using daily prices of 45 well-known cryptocurrencies between September 2019 and April 2020—the majority of which are also used in our present study. In particular, they measured the stability of cryptocurrency time series using Largest Lyapunov Exponent and Approximate Entropy. All time series are divided into two parts: the first part spans September to December 2019, considered normal time, while the second spans January to April 2020, considered a pandemic period. They revealed that the pandemic increases in cryptocurrency market uncertainty as prices fluctuated significantly. Moreover, the same experiment has also been carried out on the stock market, results indicating a lower level of price fluctuations in the stock compared to digital currencies. Also on the same topic, Drozdz et al. [21] compared the Pearson correlation between the cryptocurrency market and different asset classes including stocks, fiat currencies and commodities, revealing that these conventional markets easily influence the cryptocurrency market when they are in turbulent times, while there is no significant correlation between digital currencies and other markets in normal times, given the time resolutions they used are 10 and 360 mins.

Reactions of the general public to the COVID-19 outbreak were also observed to examine its relationship with cryptocurrencies' returns. For example, authors in [40] measured the fear of people by the frequency of occurrence of keywords COVID-19 and coronavirus on Google Trends (https://www.thinkwithgoogle.com/, accessed on 4 August 2022). Thanks to the vector autoregressive (VAR) models, they compared the evolution of this fear with the stock market's expectation of volatility VIX index (the VIX index is a measure of constant, 30-day expected volatility of the US stock market, derived from real-time, mid-quote prices of S&P500. Normally, it is calculated using the Black–Scholes formula) as well as the Bitcoin returns. They found that increases of fear can lead to Bitcoin crashes, as the correlation coefficient is -0.9. Furthermore, negative sentiment generated by coronavirus news is associated with market volatility, which is in line with other findings such as in [41]. Interestingly, some studies on the relationship between news-based sentiment and cryptocurrencies showed that, although both bad and good news cause the change in the returns and volatilities of cryptocurrencies, positive news has more effect on the volatilities and returns of cryptocurrencies in comparison with negative news [42-44].

Recently, network analysis in the cryptocurrency market during the COVID-19 pandemic has been carried out, with the common result being that the pandemic, as well as the global downturn, actually caused a change in the network structure of the cryptocurrency market. Specifically, cryptocurrencies tend to form bigger groups during the downtime, i.e., the number of potential clusters found in the network decreases during the downtime, with a few cryptocurrencies acting as central nodes. This topic has only been explored in a few studies to date [21,22,45,46]. Moreover, there are some gaps: (1) the lack of deep investigation of the network structure as they only consider MSTs; (2) the noise and trend effects are not removed; (3) data limitation issues.

We will address these shortcomings by doing deeper experiments on the network structure of the cryptocurrency market before, during and after the COVID-19 pandemic via a longer dataset with the effect of noise and trend removed. In addition, we will look at the way cryptocurrencies form a group during turbulent times by considering their rankings (identified by its market capitalization, the larger its maket capitalization, the higher its rank). We believe that this research can propose a better understanding of interconnections between digital currencies during standard and unstable periods. Furthermore, we also aim at understanding the investment decision of investors in different market states based on the results of community detection.

#### 3. Data Description

All experiments in this study have been carried out based on a tick-by-tick price dataset (tick data are the highest resolution intraday data and consist of the sequence of each executed trade or bid/ask quote aggregated from an exchange) that was collected from the hitBTC exchange (a platform for digital asset and currency exchange to quickly and securely trade cryptocurrencies—website address: https://hitbtc.com/) from 13 February 2019 to 6 April 2021. The dataset comprises 34 cryptocurrencies with a hybrid of high and low rankings. Specifically, the highest rank is 1 (Bitcoin) while the lowest rank is 260 (FunToken), according to the price-checking website *Coinmarketcap* (https://coinmarketcap. com, accessed on 4 August 2022) in April 2021; full list in Table 1.

		Cryptocurrencies			
Argur	Bitcoin SV	Ethereum Classic	MaidSafeCoin	Ontology	Tron
(RĔP)	(BSV)	(ETC)	(MAID)	(ONT)	(TRX)
Bancor	Cardano	FunToken	Maker	Ox	Verge
(BNT)	(ADA)	(FUN)	(MKR)	(ZRX)	(XVG)
Basic Attention	Decentraland	ICON	Monero	OTIM	Zcash
Token(BAT)	(MANA)	(ICX)	(XMR)	QTUM	(ZEC)
Bitcoin	Dogecoin	IOST	Nem	Ripple	Zilliqa
(BTC)	(DOGE)	1051	(XEM)	(XRP)	(ZIL)
Bitcoin Cash	FOS	Lisk	NEO	Stellar	
(BCH)	EOS	(LSK)	NEO	(XLM)	
Bitcoin Gold	Ethereum	Litecoin	OMG Network	Tezos	
(BTG)	(ETH)	(LTC)	(OMG)	(XTZ)	

Table 1. A list of 34 cryptocurrencies used in this study. Abbreviations are put in parentheses.

#### 3.1. A Note on Data Sampling and Missing Data

Since price values are collected tick-by-tick, there is no fixed timescale for all cryptocurrencies leading to an inconsistency between the time series. For this reason, we re-sample the dataset by using data points at a specific timescale. In particular, we choose four different timescales, namely 30 min, 6 h, 12 h and 24 h. Each data point of a dataset is taken to be the price of the last transaction of 34 cryptocurrencies within the considered timescale. Eventually, we have four datasets corresponding to four different timescales. Table 2 shows the description of each re-sampled dataset.

Table 2. Characteristics of four re-sampled datasets at four different levels of granularity.

Level of Granularity	# Data Points	# Missing Values
	37,632	289 (0.8%)
6 h	3136	24 (0.8%)
12 h	1568	12 (0.8%)
24 h	784	0 (0%)

Three out of four datasets have missing values with the same percentage of 0.8%. Note that a data point of a dataset is considered missing if at least one cryptocurrency does not have the price value at this data point. For each time series, instead of simply removing missing values from the time series and values from other time series from the same time, we replace missing values with the average value of the corresponding time series. This technique has been adopted in different research topics with good performance [47–49]. Furthermore, we notice that this does not change the statistical properties of the correlation between time series but, instead, helps to keep more information and thus the results found from conducting the experiments are more reliable and accurate.

#### 3.2. Aggregational Gaussianity

Aggregational Gaussianity is considered a stylized fact in traditional financial markets. In [50], the authors observed the evolution of distributions of the IBM stock returns by looking at different levels of granularity, e.g., 30 min, one day, one week and one month, finding evidence of Aggregational Gaussianity. Another study on this topic drawing the same conclusion is described in [51]. However, these authors used different stocks and a higher set of timescales from one day to one year, showing that this stylized fact is also true for stocks at coarser time resolutions.

We investigate whether Aggregational Gaussianity exists in our log-return time series using a set of four timescales: 30 min, 6 h, 12 h and 1 day. We observe this statistical aspect by implementing three experiments: Firstly, we construct the histogram as well as kernel density estimation (KDE) for each cryptocurrency time series. Secondly, we generate the Q-Q plot, which is a popular approach to test normality for a time series [52]. Lastly, we use the Lilliefors hypothesis test for normality [53]. We obtained the following findings: firstly, although the distributions of these cryptocurrency time series have a bell curve shape at all timescales considered, they are not (from the Q-Q plot and Lilliefors test) normally distributed; secondly, however, there appears to be evidence to say that Aggregational Gaussianity exists in all cryptocurrencies used in this present study from the Q-Q plots. This result is in line with existing findings in the cryptocurrency market such as [54,55].

#### 4. Research Methodology

#### 4.1. Correlation Matrix Based on Pearson Coefficients and Random Matrix Theory

Given  $x_i$  is the price time series of cryptocurrency *i*, we use its return values to find the correlation between cryptocurrencies. This is because Return values are represented as a percentage, making them scale-free and especially, stationary, which is an important requirement for many statistical tools, such as *Normalization*. Thus, we first calculate the corresponding return time series  $r_i$  as follows [56]:  $r_i = log(x_i^t/x_i^{t-1})$ , where  $x_i^t$  is the price value of the cryptocurrency *i* at timestamp *t*.

Each of these return time series can be normalized as follows [57]:  $\hat{r}_i = (r_i - \mu_i)/\sigma_i$ , where  $\mu_i$  and  $\sigma_i$  are the average value and standard deviation of time series *i*, respectively.

We form a  $m \times n$  matrix **G** such that each column represents a normalized return time series of a cryptocurrency and each row represents a timestamp. The corresponding correlation matrix **C** can be expressed as follows [56]:  $\mathbf{C} = \frac{1}{m}\mathbf{G}\mathbf{G}^{\mathsf{T}}$ . In other words, each element  $C_{ij}$  of **C** shows the correlation strength between cryptocurrencies *i* and *j* by calculating the dot product of the two normalized return time series,  $C_{ij} = \langle \hat{r}_i, \hat{r}_j \rangle$ . Such a correlation matrix is called *Pearson correlation matrix*.

It should be noted that Pearson correlation has some limitations as described in [58]. In particular, its sensitivity to outliers and inability to capture non-linear relationships both have the potential to cause misleading results. However, we believe that this correlation metric is appropriate to use in our study for the following reasons:

- Firstly, we make use of cryptocurrency returns in order to retain the statistical nature
  of the associated time series. While some authors have proposed addressing the nonlinearity problem (e.g., Spearman [59] and Kendall [53]), these have the disadvantage
  of converting rational numbers into integer rankings, with the potential to lose out on
  critical information from financial time series [60]. Moreover, it has been shown that
  rank correlation metrics also suffer from the nonlinearity issue in some cases [58].
- Secondly, Pearson has been widely applied in the existing literature, not only in the cryptocurrency market [21,22,32] but also in markets for more traditional asset classes [2,6,24]. This strongly reinforces our belief in the applicability of this method of correlation calculation for our problem.
- Thirdly, rank-based correlation metrics require independent observations. This is a known weakness of non-linear correlation methods such as Spearman and Kendall [60]. On the other hand, Pearson works well for time series with duplicate observations (because there is no requirement for independent observations), as is the case in financial time series. For example, the price of a cryptocurrency can be unchanged for a period of time.

One issue raised from this type of matrix is the question of how reliable these correlations are, in other words, whether the correlation matrix shows genuine and authentic relationships between the considered time series. Thanks to the RMT [61], this hypothesis can be examined. Particularly, given a  $m \times n$  random matrix **N** whose elements are distributed randomly with zero mean and unit variance, the eigenvalue distribution of the correlation matrix  $\mathbf{C}_{\mathbf{N}} = \frac{1}{m} \mathbf{N} \mathbf{N}^{\mathsf{T}}$  follows the Marchenko–Pastur probability density function [62] if the Quality Factor  $Q = \frac{m}{n} \ge 1$  holds when the number of timestamps  $m \to \infty$  and the number of features  $n \to \infty$ :  $P(\lambda) = \frac{Q}{2\pi} \frac{\sqrt{(\lambda + -\lambda)(\lambda - \lambda_{-})}}{\lambda}$ , where *P* is the Marchenko–Pastur probability density function,  $\lambda$  is an eigenvalue of  $C_N$ ,  $\lambda_{\pm} = 1 + \frac{1}{Q} \pm 2\sqrt{\frac{1}{Q}}$  are upper and lower limits, respectively.

From RMT, eigenvalues falling outside of  $[\lambda_-, \lambda_+]$  are assumed to deviate from its expected predictions [63,64]. Hence, we can use this theory to test the reliability of the relationships in our empirical data [65]. That is, if an empirical correlation matrix actually has real valuable information, it must have eigenvalues that are outside the bounds of  $[\lambda_-, \lambda_+]$ . Otherwise, the empirical correlation matrix can be taken to contain mainly random noise. In this study, RMT has been used to test our correlation matrices. The results show that all correlation matrices are not random and contain valuable information.

# 4.2. Cleaning Trend and Noise Effects in the Cryptocurrency Market

#### 4.2.1. Noise and Trend

The cryptocurrency market is known to have a higher percentage of noise than other traditional financial markets. According to [66], the average daily signal-to-noise ratio of the cryptocurrency market is 36%, which is extremely low compared to well-established US stock exchanges such as NYSE and NASDAQ, with an average daily signal-to-noise ratio of 90%, given the considered period between March 2017 and November 2017. The noise in the cryptocurrency market might come from different sources. For instance, there is no fixed volume for a transaction to be executed at a time, so investors can freely choose the amount that they want to trade; however, this issue causes one problem, in that investors can reduce the transaction costs by splitting their budget into smaller pieces and then buy one cryptocurrency many times with different amounts of volume and price, a practice which can trigger unforseen price movements, see [67]. Furthermore, cryptocurrencies' prices are vulnerable to "pump and dump" schemes [68], which have become pervasive recently, and also regulatory news enacted by national authorities [69]. All of these factors might intervene in the price movements of digital assets. Consequently, the correlation matrix between cryptocurrencies cannot explain their real connections as it is highly influenced by these noise factors.

On the other hand, the trend effect found in other correlated systems [70] might be found in the cryptocurrency market. Briefly speaking, a trend among cryptocurrencies means that they tend to move together in terms of price values. We notice that the majority of cryptocurrencies are created based on the protocol of leading cryptocurrencies such as Bitcoin and Ethereum (e.g., MKR, BNT, ICX, ETC and LTC) [71]. Moreover, cryptocurrencies' prices readily fluctuate with mass media [72], causing a herding behavior [72]. Similar characteristics contribute to creating a trend in cryptocurrencies.

Generally, these phenomena might be reasons for a high-value correlation matrix of cryptocurrencies from our dataset. Thus, it is important to remove of the existing noise and trend before moving on to further analysis.

#### 4.2.2. Cleaning Method

In recent studies, different approaches have been proposed to remove the noise from a correlation matrix through modification of the corresponding eigenspectrum, e.g., Linear shrinkage [73], Eigenvector clipping [74], Non-linear shrinkage [75] and Rotationally invariant, optimal shrinkage [76]. One common obstacle for most of the existing cleaning methods is that they have parameters needing definition. This raises an obvious question: how do we choose these? It is acknowledged that a lot of effort has been made to obtain the right parameter values, i.e., the noise is removed completely without the loss of data information [77,78]. However, these optimization approaches have one issue, which is that they use the Frobenius norm in their formula, so they fail to work with outlier-containing data, a downside of the Frobenius metric [79]. On the other hand, Eigenvector Clipping distinguishes itself from others [74] as it does not require any training parameters, making its outcome robust and more reliable. Furthermore, this cleaning method is straightforward to implement, with the guaranteed efficiency as it keeps the information part, i.e., after the cleaning process, the trace of the correlation matrix remains unchanged [80]. This

method has shown good performance in different studies and has been applied widely to different topics such as programming education, portfolio optimization and signal processing [70,81,82]. The outstanding performance of the Eigenvector clipping encourages us to choose this method for our cleaning scheme.

Given eigenvalues  $\lambda_1 \geq \lambda_2 \geq \lambda_3 \geq \ldots \geq \lambda_n$  and corresponding eigenvectors  $v_1, v_2, \ldots, v_n$  of our empirical correlation matrix **C**, we can identify  $k \leq n$  such that  $\lambda_k > \lambda_+$  and  $\lambda_{k+1} \leq \lambda_+$ . The Eigenvector clipping defines the denoised correlation matrix **C**<sub>denoised</sub> by [83]:

$$\mathbf{C}_{denoised} = \Sigma_{i=1}^{n} \lambda_{i}^{*} v_{i} v_{i}^{\mathsf{T}}, \lambda_{i}^{*} = \begin{cases} \frac{\lambda_{k+1} + \lambda_{k+2} + \dots + \lambda_{n}}{n-k}, \forall i \ge k+1\\ \lambda_{i}, \forall i \le k \end{cases}$$
(1)

Equation (1) uses the same eigenvectors as **C** but modifies their corresponding eigenvalues such that those greater than  $\lambda_+$  remain unchanged while the rest will be replaced by their average value. Notably, although small eigenvalues are replaced, the trace of the denoised correlation matrix is equal to its origin.

Regarding the trend effect, it is explained by the first eigenvalue and eigenvector, referred to as "market component" [83]. The market component is proved to influence the outcome of the correlation matrix. In particular, it is involved in all interactions observed from the correlation matrix due to its enormous amount of information, consequently, lessening the performance of clustering algorithms [84]. Thus, removing this component is a necessary step to clean the trend effect so that a greater portion of the correlation can be explained by components that affect specific subsets of the cryptocurrencies and, hence, facilitate clustering algorithms to find dissimilarities across clusters. A cleaned correlation matrix  $C_{cleaned}$  is obtained by subtracting the market component from the denoised correlation matrix:

$$\mathbf{C}_{cleaned} = \mathbf{C}_{denoised} - \lambda_1 v_1 v_1^{\mathsf{T}} \tag{2}$$

We found that the connections between cryptocurrencies decrease greatly without noise and trend effects: large cryptocurrencies such as Bitcoin, Ethereum and Ripple do not see to affect the cryptocurrency market as they did before the cleaning process, since there is no strong connection between them and other cryptocurrencies. This result is in line with [70], where the Eigenvalue Clipping method was also used to clean the educationrelated correlation matrix.

#### 4.3. Distance Matrix and Its Minimum Spanning Tree

Although the correlation coefficient can explain some aspects of the relationships between cryptocurrencies, it is not a metric [85]. Thus, the connections learned from the correlation matrix lack topological characteristics because they are not placed in a metric space [85]. To tackle this issue, a concept named *Distance Matrix* has been introduced to replace the correlation matrix.

Let **D** be a distance matrix deriving from  $C_{cleaned}$ , then:

$$d_{ij} = \sqrt{2 * (1 - c_{ij})}$$
(3)

where  $d_{ij} \in [0, 2]$  is an element of **D**, with 0 indicates the complete similarity between 2 nodes while 2 indicates the complete difference between 2 nodes. From the Equation (3), we can prove that: (1)  $d_{ij} \ge 0$ , (2)  $d_{ij} = 0$  if i = j and (3)  $d_{ij} = d_{ji}$ , i.e., the requirements of a metric are satisfied [85]. By using the distance matrix, we can derive a network (graph) of cryptocurrencies (nodes) with a specific topology, where similar cryptocurrencies are close to each other and cryptocurrencies with different behaviors are far away from each other, the link (edge) between each pair of cryptocurrencies is their distance value. Thanks to this topology, different communities of cryptocurrencies can be observed.

One problem with this type of network is that it is dense. That is, for a set of N nodes, the corresponding graph deriving from **D** has  $\frac{N \times (N-1)}{2}$  edges such that each vertex connects to all other vertices. To reduce the complexity of the network, we use a Minimum Spanning Tree (MST) [86], which refers to a special tree from the graph that links all vertices together in which its length is minimal. Particularly, it reduces the amount of redundant information since it only keeps the N - 1 most important edges, i.e., N - 1 shortest edges that are well connected. MST stems from graph theory and is applied widely to different fields [4,87,88], especially in financial markets [89–91]. To exploit the useability of MST, the dynamics of community structures in the stock market are observed by Huang et al. [92] with the dataset split into consecutive smaller periods and a MST constructed at each of them. Thus, the characteristics of a financial network can be captured by observing the evolution of MSTs. More recently, the cryptocurrency market was introduced and attracted a number of investors, and the demand for exploring the correlation between cryptocurrencies thereby emerged. However, this topic is rather new and needs more studies to be implemented [4,93].

There are two famous algorithms to find the MST, namely Prim [93] and Kruskal [94]. While both methods show good performance, Kruskal seems to be better in terms of time complexity. A comparison between the two from [95] shows that the prior works well with a big network, while the latter is dominant when the network is small, which is appropriate for this study as we have only 34 cryptocurrencies. Moreover, Kruskal is used more often in finance-related topics compared to other approaches [96–98], which strengthens the reliability of the algorithm. With these advantages, we choose Kruskal for this study.

#### 4.4. Community Detection in the Cryptocurrency Market

Given a MST from the distance matrix **D**, different communities are formed and can be recognized clearly, i.e., cryptocurrencies belonging to one community have short distance edges among them and the distance between two others in two different communities is longer than any edges of these two communities. However, there are less common cases in which some nodes are scattered between communities, or it is not visible from the graph how close the two communities are. This issue motivates us to further analyze the MST to optimize the clustering result using several community detection methods which have been developed [99–103]. Of these, the Louvain method is applicable across a wide range of domains [104–107]. Thus, we apply this method to our MST in order to obtain optimal communities. Theoretically, Louvain is an optimization problem that uses *Modularity* to measure the density of links inside communities compared to links between communities. The target of Louvain is to minimize the Modularity measure, which means that different authentic communities are clustered very tightly [108].

However, it is not convincing just to show results from one method only, as the community structure of a network might be just random. To overcome this issue, we also adopt another commonly used method named Girvan–Newman, which removes edges from the original graph one-by-one such that the edge having the highest number of shortest paths between nodes passing through it is removed first. Eventually, the graph breaks down into smaller pieces, so-called communities [109].

If the results proposed by these two community detection methods are similar, it implies that the relationship of the cryptocurrencies as well as their corresponding community structure are reliable and reflects their genuine characteristics. The results after applying these methods are shown in Section 5.

#### 4.5. Time Window Division

Given the dataset described earlier, one important question about constructing a network structure in these cryptocurrencies is how to split the dataset into different consecutive periods. This is because a network structure corresponding to each period of time should be able to explain what has happened to the cryptocurrencies throughout that time, i.e., there must be a reason behind this topological structure. If we divided the dataset randomly, we could not capture important historical events at a specific period. As a result, the topology we found would be meaningless in the corresponding time window. To this end, we must select time windows rationally. We note that our dataset contains the period of the COVID-19 pandemic as well as the global downturn 2020. From the literature in Section 2, we see these historical events actually adversely influenced the financial markets. Thus, we postulate that the COVID-19 pandemic is a reasonable milestone to separate our dataset.

To verify the pandemic's impact on the global economy and thereby choose the right time windows for the dataset, we consider the movements of four different factors. Firstly, the attention to the COVID-19 pandemic, as measured by the frequency of COVID-19-related keywords searched on Google Trends. For this factor, we use two keywords including *COVID-19* and *coronavirus disease 19*. Secondly, we use the VIX index to observe fluctuations of the stock market, this index starts at 0 for no upper bound and a higher value implies that the stock market has stronger fluctuation. Thirdly, we also observe the prices of the *S*&*P*500 index, representing the US economy. Lastly, the growth rate of the world's GDP is used as a proxy for the development of the global economy in general.

Figure 1 visualizes these aforementioned factors. From Figure 1a, people started to worry about this disease in January 2020. However, it was not until March 2020 that the COVID-19 pandemic actually caught the attention of people worldwide, as the volume of searches for COVID-19-related terms quickly peaked. This remained a topic of interest until July 2020. Furthermore, March 2020 was the month in which a pandemic-induced economic recession first occurred, seriously affecting the economy of nations worldwide. This effect is shown in Figure 1b–d. In particular, the GDP's growth rate decreased by 3.3% in 2020, which is the highest decrease ever, even worse than the Great Recession in 2007–2009 [110]. Simultaneously, the stock market fluctuated dramatically, which can be seen via the VIX index and the *S&P*500 index, both of which experienced a significant fall during March 2020. However, the economy started to recover afterward, the stock market became less fluctuated and the *S&P*500 index regained its original pre-pandemic value in July 2020.



**Figure 1.** The reaction of general public and global economy to the COVID-19 pandemic. Four factors are considered: (a) Worldwide attention to the pandemic, (b) Global GDP growth, (c) VIX index, (d) *S&P*500 index.

Consequently, we split the 784 days from 13 February 2019 to 6 April 2021 into three time windows which correspond to three different stages, including normal time, downturn time and recovery time. The details for these time windows are shown in Table 3.

**Table 3.** Three time windows used in this work (time windows split to take into consideration the COVID-19 pandemic).

Time Window	Stage	Time Span	# Days
1	Normal time	13 February 2019–31 December 2019	322 days
2	Downturn time	1 January 2020–30 June 2020	182 days
3	Recovery time	1 July 2020–6 April 2021	280 days

#### 5. Experimental Results and Discussion

This section sets out our three research questions. We will first examine the impact of noise and trend effects on the correlation between cryptocurrencies as well as their corresponding topological structure. Then, we observe the evolution of the structure according to the levels of granularity. Finally, the results from these two experiments will be used to construct the right network structure. Consequently, the corresponding community structure is identified, which is used to learn the investment decisions of crypto investors during the COVID-19 pandemic.

We note that all calculations in our study are implemented using *Python* programming language (version 3.7.14, designed by Guido van Rossum, Centrum Wiskunde & Informatica (CWI), The Netherlands). Regarding network-related calculations (e.g., network construction and network-involved metrics), we utilize the *networkx* (https://networkx.org/) package incorporated into Python.

#### 5.1. The Response of Network Structures to Noise and Trend Effects

Given the fact that there are noise and trends in the cryptocurrency market, we examine whether these factors affect the cryptocurrency network structure. Since we have four datasets corresponding to four timescales (e.g., 30 min, 6 h, 12 h and 24 h), we use both metric-related methods and visualization for all available datasets to discover the discrepancy between original and cleaned (after removing noise and trends) datasets.

To show the difference between two network structures, we choose two such metrics to measure the connection strength in a network of cryptocurrencies:

• Residuality Coefficient [93]: This compares the relative strength of the connections above and below a threshold distance value. In this experiment, we use the highest distance value ensuring connectivity of the MST as the threshold, denoted *L*:

$$R = \frac{\sum_{[d_{ij} > L]} d_{ij}^{-1}}{\sum_{[d_{ij} \le L]} d_{ij}^{-1}} \tag{4}$$

• MST-based mean distance [111]: this calculates the average distance of the MST:

$$M = \frac{1}{N-1} \Sigma_{d_{ij} \in MST} d_{ij} \tag{5}$$

An increase in these means that cryptocurrencies are further from each other. By contrast, cryptocurrencies are closer to each other if these metrics decrease. Note that although both metrics are used to examine the connection strength of cryptocurrencies, the Residuality coefficient is known to be more vulnerable to the links between cryptocurrencies in different groups, i.e., if the connection strength between cryptocurrencies in different groups increases, the Residuality coefficient will decrease dramatically, and vice versa; the connections between cryptocurrencies within one group do not affect the Residuality coefficient much [112]. On the other hand, Mean distance is more vulnerable to the links between cryptocurrencies belonging to one group, as it mainly uses the connections within a group to find the average value and ignores the connections between different groups [111].

Table 4 shows the results of the two metrics using different levels of granularity. It is clear that both Residuality coefficients and Mean distance values increase significantly when the effects of noise and trend are dismissed. This phenomenon remains unchanged in different timescales, implying that this is a genuine characteristic of the cryptocurrency market. Furthermore, a visualization of network structures before and after cleaning is shown in Figure 2 to reinforce our finding. As can be seen, the topological structure changes after the noise and trends are removed. Moreover, what happens in each time window is that the number of communities decreases after removing these effects. From these figures and illustrations, we can conclude that the connections between cryptocurrencies are caused mainly by the noise and trend effects. That is, these factors result in different cryptocurrencies becoming closer to each other and forming a group. This phenomenon can be explained by low values for Residuality coefficients and Mean distance values in the original data compared to the cleaned data. A value less than unity of the prior metric means that there are few connections greater than the threshold L. Moreover, a small value of the latter metric means that cryptocurrencies within a group are closer to each other. In summary, each group of the network is compact with strong links inside, which helps the community detection algorithm to easily cluster them. In other words, the difference between different groups is clear because the links between different groups are weak, i.e., the ones greater than L. However, after cleaning the correlation matrix, cryptocurrencies that are closely related to each other through noise and trend become further away, i.e., the strong links between some cryptocurrencies are broken. This causes our metrics to increase dramatically, which means that the network structure starts to expand, forming a sparse network. For example, the Residuality coefficient of the second time window in the 30 min original data is 0.28, while it is 20 times higher after cleaning the effects of noise and trends. This fact is also true for the rest of our dataset. The result is in line with [20]; these authors did not consider the noise effect but, with the removal of trends, they found that the correlation between the 80 most liquid cryptocurrencies from 1 January 2020 to 1 October 2021 decreased.

**Table 4.** Cryptocurrency network connection strength through three time windows measured by Residuality Coefficient and Mean Distance. Four different granularity levels are considered, each with datasets, including original and cleaned dataset after removing noise and trend effects.

Matric	Data Type	Time	Granularity			
wiethe	Data Type	Window	30 min	6 h	12 h	24 h
		1	0.41	0.11	0.16	0.08
	Original Data	2	0.28	0.111	0.06	0.05
Residuality	Oliginal Data	3	0.14	0.05	0.07	0.34
Coefficient		1	1.69	6.66	14.82	14.40
coefficient	Cleaned data	2	5.98	8.90	14.41	15.34
		3	2.32	2.99	1.88	1.05
		1	1.08	0.82	0.80	0.76
	Original Data	2	0.99	0.71	0.65	0.56
Moon	Oliginal Data	3	0.98	0.57	0.46	0.45
distance		1	1.29	1.38	1.42	1.42
aistance	Closped data	2	1.40	1.42	1.42	1.42
	Cicaned data	3	1.29	1.12	1.01	1.22



(e) Time window 3, original

(**f**) Time window 3, cleaned

**Figure 2.** Cryptocurrency network structures using daily data. For each time window, Louvain method is applied to both original and cleaned data to detect existing communities. The illustrations on the left and right hand side are for the original and cleaned data, respectively, for 3 time windows referring to normal, downturn and recovery times, respectively.

# 5.2. Real Network Structures in Different Levels of Granularity: An Experiment on Cleaned Data

In this section, we will construct the network structure of 34 cryptocurrencies removing the effects of noise and trends. By doing this, we can look at the evolution of network

structures at each timescale over time and, of greater interest, the differences in the network structures between different timescales. Note that community detection results found by using Louvain algorithm are also included in these networks. The results of this experiment can shed light on the influence of timescales on cryptocurrencies' connections and what timescale should be used for cryptocurrency-related analysis.

#### 5.2.1. The Evolution of the Cryptocurrency Network According to Timescales

Figures 3–5 show the results of network structures along with detected communities using the Louvain method with each figure representing a different time window. For each window, four network structures corresponding to four different levels of granularity are displayed. One obvious statement that can be made from the illustrations is that the community structures at each level of granularity change over time. Additionally, if we consider different levels of granularity at the same time, the number of detected communities tends to decrease when the timescale becomes more coarse-grained. For large timescales, such as 24 h, cryptocurrencies build up big groups with few cryptocurrencies acting as central nodes that link directly to the remainder. For example, in Figure 3d, MANA acts as a central node that links all other cryptocurrencies together. This explains why community detection techniques cannot distinguish several subsets as the network in this case is naturally one group. Figure 4d shows a similar pattern, while in Figure 5d there are two central nodes that create two big groups with relatively similar sizes. To this end, with low-frequency data, we expect we can predict the long-term trend of cryptocurrencies in the future by looking at the central nodes from their corresponding community structures. If this is the case, it will be very beneficial for investors who choose a long-term investment. However, this behaviour requires deeper investigation and will be the subject of further research.

![](_page_14_Figure_4.jpeg)

(a) Time window 1, 30 min (b) Time window 1, 6 h

![](_page_14_Figure_6.jpeg)

(c) Time window 1, 12 h (d) Time window 1, 24 h

**Figure 3.** Network structure for the first time window, community detection is applied using Louvain method. Four different timescales are used, e.g., (**a**) 30 min, (**b**) 6 h, (**c**) 12 h, (**d**) 24 h.

We notice that the difficulty of detecting communities in this market increases with the timescale length. In other words, cryptocurrencies are more likely to belong to the same community if we just look at their price values at a high level of granularity such as daily. Thankfully, it can be explained based on the nature of the cryptocurrency market. In particular, the cryptocurrency market is well-known for its high volatility compared to other traditional asset classes such as stocks, bonds and commodities [113–116]. In [117], the authors used 5 min data of Bitcoin prices traded on three different exchanges, Kraken, Bitstamp and Btcbox, during the period between 2017 and 2021 to calculate the realized volatility (the assessment of variation in returns for an asset by analyzing its historical returns within a defined time period) of this most stable and popular cryptocurrency. The results showed that although Bitcoin is the most valuable and trustworthy cryptocurrency, its volatility fluctuates from 4.8 to 7.5. By contrast, with the same level of granularity, the stock market seems to be more stable, as the realized volatility stood at roughly 0.58 during normal times [118] and increased to just around 1.0 during the COVID-19 pandemic [119]. These facts suggest that the cryptocurrency price fluctuations are dramatic even within a 5 min period. Consequently, using a low-frequency dataset such as 12 h or 24 h appears to cause a loss of important information that influences the results of analysis. This problem has also been described in earlier studies such as [12]. However, existing studies mainly

focused on daily data to detect communities in the cryptocurrency market.

![](_page_15_Figure_2.jpeg)

(a) Time window 2, 30 min (b) Time window 2, 6 h

![](_page_15_Figure_4.jpeg)

(c) Time window 2, 12 h (d) Time window 2, 24 h

**Figure 4.** Network structure for the second time window, community detection is applied using Louvain method. Four different timescales are used, e.g., (**a**) 30 min, (**b**) 6 h, (**c**) 12 h, (**d**) 24 h.

In this study, the loss of information by using large timescales including 6 h, 12 h and 24 h makes judging the correlation between different cryptocurrencies unclear. As a result, it affects the corresponding MST which can be seen in Figures 3–5. Ideally, we would like to use a dataset that is as fine-grained as possible. Unfortunately, our experiments show that for frequencies lower than 30 min, there are a huge amount of missing values as some cryptocurrencies are not traded frequently [120], thus requiring their removal or imputing a value. This adversely affects the correlation between time series and impacts our analysis. Finally, we choose a 30 min dataset for further experiments.

![](_page_16_Figure_1.jpeg)

(a) Time window 3, 30 min (b) Time window 3, 6 h

![](_page_16_Figure_3.jpeg)

(c) Time window 3, 12 h (d) Time window 3, 24 h

**Figure 5.** Network structure for the third time window, community detection is applied using Louvain method. Four different timescales are used, e.g., (a) 30 min, (b) 6 h, (c) 12 h, (d) 24 h.

#### 5.2.2. Louvain vs. Girvan–Newman for Community Structure Detection

The Louvain method is our main technique for detecting communities but we also use the Girvan–Newman method to double-check the communities found. The *v-measure* gives the similarity between these two methods [121], shown in Table 5. This metric ranges from 0 to 1 such that 0 indicates a complete dissimilarity between two graphs while 1 indicates a complete similarity. We found that the *v-measure* in all cases is high with the lowest value of 0.82 from the third time window in the 6 h dataset in Table 5. That is, the Louvain method proposes similar results as Girvan–Newman. Thus, the communities found by Louvain are reliable for use in further analysis.

	Granularity			
	30 min	6 h	12 h	24 h
Time window 1	0.88	1.00	1.00	1.00
Time window 2	1.00	1.00	1.00	1.00
Time window 3	0.87	0.82	0.91	1.00

Table 5. *v-Measure* between Louvain and Girvan–Newman methods.

5.3. Analysis of Investors' Investment Decisions Based on the Time-Varying Network Structure 5.3.1. The Changes in Crypto Network Structure during Times of Crisis

To observe the growth of the network structure over time, we use *Degree Assortativity Coefficient* [122] and *Average Betweenness Centrality* [3]. However, these metrics fail to tell us the similarity between two networks. Thus, to statistically compare the topological change between two networks, we use three more metrics, including *v-measure*, *Degree centrality* [26] and *Eigenvalue method* [123,124].

Table 6 shows results of *Betweenness Centrality* and *Degree Assortativity*. Immediately, we can see that there is a huge change occurring in time window 2, which corresponds to the turbulent time caused by the pandemic on both metrics.

Regarding the *Betweenness Centrality*, this metric decreases from 0.15 in time window 1 to 0.05 in the next period before going back to its original value prior to the pandemic outbreak (time window 1). A reasonable explanation for this movement is that the network structure of the cryptocurency market during normal times appears to have a dispersive tendency with the whole network divided into multiple small-size groups such that each group share common characteristics. However, during COVID-19, these synchronize to form a big group. Thus, the number of groups decreases while the size of each group increases. This might be a consequence of an increase in the connectedness of cryptocurrencies during the pandemic, as shown in many research papers [11,27,45]. In the recovery time, however, the network started to divide into smaller parts again, perhaps because the cryptocurrency market overcame the most connected period and started to go back to its normal behavior.

The *Degree Assortativity* results strongly support those of the *Betweenness Centrality*. In particular, a negative value shows that high-degree nodes are more likely to link to low-degree nodes, which means that each group in the network has one node acting as a central node connecting to the rest. While the values in time window 1 and 3 are approximately the same, time window 2 shows a decline by nearly 50 percent. This indicates that the number of connections between high-degree nodes and low-degree nodes increases, i.e., the network forms big groups with a large number of leaf nodes in each group.

We notice that this time-varying structure is similar to what have been shown in works of Drozdz et al. [20,21], who stated that the market has a distributed-network topology or a hierarchical-network topology in which no node dominates the network during normal times. However, it becomes more centralized during the pandemic and started to distribute as this turbulent time is gone. More recently, another work proposed by Nie also confirmed the same result [22].

Table 7 shows results of the three similarity metrics for different time periods: *normal time* (time window 1), *downtime* (time window 2) and *recovery time* (time window 3). Each values shows the similarity between two time windows. For *v-measure*, the higher the value is, the more alike two networks are. On the other hand, for the remaining values, a lower value indicates that two networks are more similar.

The differences between time window 2 and the other two time windows are very clear. In particular, the *v-measure* between time window 1 and 3 is 0.32, meaning that communities found in time window 3 hold roughly one third of characteristics from time window 1's communities. By contrast, *v-measure* values between time window 1 and 2 as well as between time window 2 and 3 are negligible, standing at 0.04 and 0.02, respectively. Additionally, for the topological structures of MSTs, the other two metrics also show the same principle since time window 1 and 3 share common characteristics and the similarity degree of other cases are nearly zero. Remarkably, the *Eigenvalue method* shows a significant divergence of time window 2 with others, as shown in Table 7.

The severe pandemic and the global downturn of March 2020 together seem to have actually changed the way cryptocurrencies interact with each other. The changes of these interactions have created new communities and broken down old ones, i.e., some cryptocurrencies become closer to each other while others moved further away from each other due to the COVID-19 pandemic and the economic recession. Eventually, the topological structure during this turbulent time shows completely different patterns compared to the periods when the global market is stable. Furthermore, we noticed that the community structure started to recover back to its pre-COVID-19 levels after June 2020, which coincides with the time the global economy recovered and the COVID-19 pandemic had less impact. During this time, some characteristics of the network structure reappeared that are similar to the structure during the normal time (it is obvious that these structures are not fully similar because they change over time, as proven in previous sections and, in addition, after June 2020, the global economy started to recover, but not as well as in the past, and the pandemic still had an impact on the economy worldwide to some extent). This is why the *v-measure* between time window 1 and 3 increased significantly and the corre-

sponding differences measured by *Degree centrality* and *Eigenvalue method* are very small. The community structures for the three time windows are shown in Figures 3a, 4a and 5a.

**Table 6.** The growth of network structures over time measured by Betweenness Centrality and Degree Assortativity.

Metrics	Time Window 1	Time Window 2	Time Window 3
Betweenness centrality	0.15	0.05	0.16
Degree Assortativity	-0.49	-0.72	-0.51

**Table 7.** Similarity in network structures between different phases of the cryptocurrency market measured by three metrics. A higher value of *v*-measure indicates a greater similarity between two structures, whereas, higher values of *degree centrality* and *eigenvalue method* indicate more dissimilarity between two structures.

Time Window	1 vs. 2	1 vs. 3	2 vs. 3
Metrics			
Degree centrality	0.5	0.09	0.42
Eigenvalue method	844.45	4.59	759.16
<i>v</i> -measure	0.04	0.32	0.02

5.3.2. Learning the Investment Decision of Crypto Traders Based on Ranking Distribution

The ranking of a cryptocurrency is measured by its market capitalization (a multiplication between the number of coins in circulation and the current market price of a single coin). We obtain cryptocurrencies' ranking on the https://coinmarketcap.com website (accessed on 15 August 2022).

We use this characteristic of cryptocurrencies to examine how they are distributed in each community of the cryptocurrency network. More importantly, we will have a look at the way cryptocurrencies form groups during different phases of the global economy by observing the distribution of ranking in each group between different periods of time.

Table 8 summarizes the results of community detection using the Louvain method. For each period of time, the found communities are listed with a set of cryptocurrencies and corresponding rankings belonging to each of them. We found that during the normal time, there is a mix between high-ranking and low-ranking cryptocurrencies in each community. For example, group 6 has a size of 7 including top-ranking cryptocurrencies such as BTC, ETH and BCH, while also having very low-ranking ones such as MAID and ICX. We pay more attention to communities found in the downturn time. At this phase, we recognized that the community formation of these cryptocurrencies seems to be dramatically different from the previous period. In particular, there are only two communities found during this period, while the other has six. More importantly, there seems to be a separation between high-ranking and low-ranking cryptocurrencies, because the majority of top-ranking cryptocurrencies belong to one group while the majority of low-ranking cryptocurrencies are in the other. Additionally, by comparing these results with the period of recovery, we noticed that this period shares common characteristics with both normal time and downturn time. Specifically, after the downturn time, cryptocurrencies started to separate from each other; this can be seen by looking at the number of communities during this time. There was an increase from 2 to 6, which is equal to the normal time case. While the majority of communities show a mix between high- and low-ranking cryptocurrencies, there are two communities that are similar to the downturn time: group 4 with all high-ranking cryptocurrencies and group 5 with all low-ranking cryptocurrencies.

	Group	Cryptocurrencies	Rankings
	1	ADA, XLM, BAT, ZIL	10, 13, 32, 99
me	2	BTG, IOST, XTZ, ZRX, ETC	12, 21, 45, 57, 83
al ti	3	LSK, OMG, REP, FUN, MKR	26,54, 58, 70, 168
rmő	4	NEO, MANA, BNT, XVG, XEM, QTUM	19, 31, 41, 86, 117, <b>184</b>
Ž	5	ONT, ZEC, XMR, XRP, EOS, TRX, LTC	3, 6, 7, 11, 16, 29, 35
	6	ICX, MAID, DOGE, BTC, BSV, ETH, BCH	1, 2, 5, 9, 34, 84, 130
	1	DOGE, ICX, BNT, MANA, ZRX, FUN,	32, 33, 40, 45, 60, 81,
me	L	MAID, BAT, XVG, ONT	105, 124, 139, <b>196</b>
n ti		ADA, BCH, BSV, BTC, BTG, EOS,	1 2 4 5 6 7 0
tur	2	ETH, ETC, IOST, LSK, LTC,	1, 2, 4, 5, 0, 7, 9,
им		MKR, NEO, OMG, QTUM, REP, TRX,	11, 12, 13, 17, 16,
Do		XEM, XLM, XMR, XRP, XTZ, ZEC,	21, 22, 27, 30, 34, 48,
		ZIL	51, 53, 54, 62, 65, 91
0)	1	BTG, MANA, BAT, ZEC	56, 62, 67, 107
ime	2	ONT, QTUM, EOS, BSV, MKR	24, 31, 53, 75, 88
ry t	3	XVG ,ZIL, XEM, MAID, BTC, ETH	1, 2, 38, 48, 109, 136
ove	4	ADA, DOGE, XRP, BCH, XLM, LTC	6, 7, 9, 15, 16, 20
lect	5	OMG, BNT, IOST, REP, ICX, LSK	68, 78, 85, 100, 101, <b>140</b>
R	6	ETC, ZRX, TRX, NEO, XMR, FUN, XTZ	17, 27, 33, 35, 64, 76, 129

**Table 8.** Distributions of rankings in each community during different phases of the financial market: normal time, downturn time and recovery time. The rankings are sorted in ascending order. Bold values are minimum and maximum ranks in each period.

Figure 6 shows the distribution of cryptocurrencies' rankings in three different phases of time. We use this visualization to show readers the changes of ranking distributions in a clearer and easier manner. Each community is represented by a circular shape while the rankings of cryptocurrencies are represented by the intensity of the blue color, i.e., the darker the blue, the lower the cryptocurrency's rank. Figure 6b shows that the circular shape of group 1 is clearly darker than that of group 2. On the other hand, there is a combination of both bright and dark blue in the majority of cases in two remaining sub-figures. Notably, Groups 3 and 5 in Figure 6c show a clear difference from the rest.

When it comes to these results, investors' investment decisions can be considered as potential explanations for the time-varying community structure. During normal times, i.e., when the financial market is stable and there is no major event occurring that impacts wider society, investors show a non-herding behaviour. That is, their decision for investing in a cryptocurrency is based on their own market analysis and is not influenced by other investors' choice. This might push up the vibrancy of the cryptocurrency market where a large number of coins with both high and low rankings are traded. As a result, there is a diversification in terms of rankings in each community. Empirically, it is found that there was no herding behavior before the pandemic. In particular, Larisa et al. in [125] used hourly price time series of multiple exchanges such as Binance, Bitbay, BitFinex, Coinbase and major cryptocurrencies including BTC, LTC and ETH to find the existence of herding before the start of COVID-19. Based on the Cross Sectional Absolute Deviation model, they found that the herding behavior was free during this time. By contrast, during a turbulent time, investors are panicked by the fluctuations of cryptocurrencies' price as well as being bombarded by bad news that strongly affect their investment. Different studies have been carried out to investigate the investors' behavior since the onset of the COVID-19 outbreak. Generally speaking, these reached the same conclusions: that the pandemic actually increased herding behavior in the cryptocurrency market. In [126], the authors used 43 cryptocurrencies with large market capitalization between 2013 and 2020; they

found that investors in the cryptocurrency market follow the consensus and the impact of coronavirus media coverage is significant on the herding behavior. In particular, news related to the coronavirus increases fear and affects the behavior of investors reducing appetite for risk. Consequently, investors disregard their private information and follow others' investment decisions. However, the impact of media is reduced when the market returns to a normal phase. This is in line with different studies that use different datasets and time periods [125,127,128]. More importantly, the way investors show herding behavior is that they tended to invest in the major and most-tradable cryptocurrencies [27]. This can be explained by the fact that high-ranking cryptocurrencies are more mature so they are more stable than the rest and are more likely to retain value under the uncertainty of the global financial market, causing a bias from investors [129]. Consequently, major cryptocurrencies were seen to increase in terms of trading volume and act as a store of value during the turbulence times [130]. In other words, there was a risk aversion occurring after the pandemic outburst as described in [72]. Eventually, all high-ranking cryptocurrencies belonged to one group.

![](_page_20_Figure_2.jpeg)

(c) Time window 3

**Figure 6.** Cryptocurrency's rankings distributions in three different phases of time. Each community is represented by a circular shape while the rankings of cryptocurrencies in this community are given by the blue color intensity, i.e., the darker the blue, the lower the cryptocurrency's rank.

When it comes to low-ranking cryptocurrencies, we notice that cryptocurrencies with the lowest rankings in our dataset belong to another group. This might be because they receive the same treatment from investors during the downturn time, so they have the same trend. One possible reason for this is that low-ranking cryptocurrencies are less likely to be used as an investment option during the downturn time because they have negligible value and bring more risk to investors. Instead, they are mainly used for other purposes, such as paying transaction fees, as currency for a smart contract or simply a token on a cryptocurrency platform used to access applications [27]. This seems reasonable as the pandemic stopped in-person interactions. Hence, they had to complete all work remotely. In this case, cryptocurrencies and blockchain technology are extremely useful since they thrive under the proposed online environment to resume working activities worldwide and also bring benefits to users. Being used for the same purpose causes a similarity between these cryptocurrencies.

All findings that we have shown earlier help us to explain the community structure in time window 3, which corresponds to the recovery period. During this time, the concerns about the pandemic started to decrease, meaning that not only cryptocurrency but also other traditional assets recovered with investors' newfound positive attitude bringing them back to normal trading. Crypto traders started to diversify their portfolio by investing in different low and high market-capitalized assets and making their own decisions [126]. However, one remarkable phenomenon that is worth taking into consideration is ] risk-taking behavior. A piece of research implemented by Christoph et al. [131] used 100 return time series of risky stocks to conduct a survey related to the investment behavior of professional market traders. The responses of more than 800 participants revealed that a number of investors underestimate risk after prolonged exposure to high risk, as they become accustomed to the uncertainty of the economy. Thus, they go back to investing in risky assets or even engage in more risk-taking to gain more profits. This tendency explains the similarity in the community structures between time windows 1 and 3. However, as we can see, there exists one group with high-ranking cryptocurrencies and one group with low-ranking ones as a result of the risk aversion of a portion of investors after the great shock caused by the pandemic.

#### 6. Limitations and Future Works

#### 6.1. Limitations

Although the tick-by-tick dataset used in this study is large, which strengthens the results of the experiments, the number of cryptocurrencies should ideally be higher so that we can draw firmer conclusions regarding the cryptocurrency market (e.g., whether the results generalise for large-cap and small-cap crypto assets). This will be the subject of future work. Secondly, while 30 min granularity has been found to suffice for our calculations, it would be better if we could use a lower level, say 15 min or even finer. Unfortunately, some cryptocurrencies are not traded regularly causing a lot of missing values at these timescales. This will also be the subject of future work.

There is also a concern with respect to the use of Pearson correlation for clustering problems. In particular, although this correlation metric has been applied widely in the existing literature and proposed various findings in the financial markets [2,21,22], it is sensitive to outliers [58] and cannot capture non-linear relationships that might cause misleading results [25]. Consequently, this adversely affects the clustering results. Indeed, these issues are also observed in other correlation metrics such as Spearman and Kendall [58]. Furthermore, we noticed that the results of clustering vary significantly by using different correlation measuring methods. Thus, it is necessary to deeply investigate different methods for a specific research task and analyze the results from each of these methods. Indeed, the creation of new approaches for calculating correlation coefficients that overcome the current limitations needs to receive more attention.

# 6.2. Future Works

Understanding how cryptocurrencies are correlated with each other sheds light on portfolio optimization. Based on the outcome of this study, we can take one step further by constructing and comparing the portfolio optimizations at different phases of the market,

i.e., during bear and bull market periods. Therefore, the unique characteristics of an optimized portfolio at different market phases can, in theory, be learned and analysed. Secondly, we have noted that different network structures can be observed for a number of exchanges. Thereby, a comparison between them can be made. Another future plan which is worth taking into consideration is to observe the correlation using different techniques. For instance, we are aiming to use mutual information, which is successfully applied in [25], to estimate the correlation between two cryptocurrencies. This method can overcome obstacles from popular linear and non-linear methods since it can measure the non-linear correlation while allowing the existence of non-monotonic relationships. Lastly, we have noticed that the network structure of low-frequency data behaves differently to that of high-frequency data. We remark that we can expect to learn the long-term characteristics of cryptocurrencies based on this structure which could be potentially beneficial for investors who choose to make a long-term investment decision.

#### 7. Conclusions

This research aims at answering three questions related to cross-correlations in the cryptocurrency market: Firstly, how do noise and trends in cryptocurrencies influence their cross-correlations and then the corresponding network structure? Secondly, what level of granularity should we use? Lastly, is the dramatic change in the cryptocurrency network structure during the pandemic caused by investors' investment strategy? We firstly analyze the effect of noise and trend in cryptocurrencies on their cross-correlations and then remove these factors thanks to Random Matrix Theory and Market Component. Four sub-datasets with different levels of granularity including 30 min, 6 h, 12 h and 1 day are created from the original tick-by-tick data to examine the importance of choosing the right frequency resolution. Then, we use MST to construct a correlation-based network and detect different potential communities by using Louvain and Girvan-Newman algorithms. We found that the correlations between cryptocurrencies are mainly caused by noise and trend effects, which might lead to a big problem for the traders' investment strategy because investors might be fooled by looking at the counterfeit relationship. It is necessary to analyze and explore real interactions between cryptocurrencies so that the evolution of the cryptocurrency market can be learned properly and thus investors can choose a good strategy for their investment. Moreover, the frequency resolution of our data plays an important role in the performance of correlation matrix and also community detection. Specifically, the finer the data, the more precise the community structure. Thus, we use a 30 min dataset, which is the finest available timescale in this study. The dramatic change in the community structures between bearish and bullish markets reveals a change in the investment decisions of investors. In particular, investors makes their own investment decisions based on their personal market analysis and experience during normal times. Eventually, this causes a diversification in the cryptocurrencies chosen to invest in, since not only high- but also low-ranking cryptocurrencies are added in the portfolios. On the other hand, investors tend to only trade cryptocurrencies with high market capitalization during turbulent times, while smaller cryptocurrencies are mainly used for other purposes, such as transaction fees, smart contracts tokens or simply used to run a digital platform.

Author Contributions: A.P.N.N.: Conceptualization, Methodology, Software, Formal analysis, Investigation, Writing—original draft, Visualization. T.T.M.: Methodology, Writing—review and editing, Supervision. M.B.: Conceptualization, Methodology, Data curation, Writing—review and editing, Supervision. M.C.: Conceptualization, Methodology, Data curation, Writing—review and editing, Supervision. All authors have read and agreed to the published version of the manuscript.

**Funding:** The APC was funded by Science Foundation Ireland Centre for Research Training in Artificial Intelligence under grant number 18/CRT/6223.

Institutional Review Board Statement: Not applicable.

Data Availability Statement: Not applicable.

Acknowledgments: The authors Martin Crane, Marija Bezbradica, Tai Tan Mai, wish to acknowledge the support, in part, from the Science Foundation Ireland under Grant Agreement No. 13/RC/2106\_P2 at the ADAPT SFI Research Centre at DCU. ADAPT, the SFI Research Centre for AI-Driven Digital Content Technology, is funded by the Science Foundation Ireland through the SFI Research Centres Programme. The author An P. N. Nguyen wishes to acknowledge the support from Dublin City University's Research Committee and research grants from Science Foundation Ireland Centre for Research Training in Artificial Intelligence under grant number 18/CRT/6223.

Conflicts of Interest: The authors declare no conflict of interest.

#### References

- Li, Y.; Jiang, X.F.; Tian, Y.; Li, S.P.; Zheng, B. Portfolio optimization based on network topology. *Physica A* 2019, 515, 671–681. [CrossRef]
- 2. Zhao, L.; Li, W.; Cai, X. Structure and dynamics of stock market in times of crisis. Phys. Lett. A 2016, 380, 654–666. [CrossRef]
- 3. Balcilar, M.; Ozdemir, H.; Agan, B. Effects of COVID-19 on cryptocurrency and emerging market connectedness: Empirical evidence from quantile, frequency, and lasso networks. *Physica A* **2022**, *604*, 127885. [CrossRef]
- 4. Chaudhari, H.; Crane, M. Cross-correlation dynamics and community structures of cryptocurrencies. *J. Comput. Sci.* 2020, 44, 101130. [CrossRef]
- 5. Wątorek, M.; Drożdż, S.; Kwapień, J.; Minati, L.; Oświęcimka, P.; Stanuszek, M. Multiscale characteristics of the emerging global cryptocurrency market. *Phys. Rep.* **2021**, *901*, 1–82. [CrossRef]
- 6. Nie, C.X. A network-based method for detecting critical events of correlation dynamics in financial markets. *Europhys. Lett.* 2020, 131, 50001. [CrossRef]
- 7. Papadimitriou, T.; Gogas, P.; Gkatzoglou, F. The evolution of the cryptocurrencies market: A complex networks approach. *J. Comput. Appl. Math.* **2020**, *376*, 112831. [CrossRef]
- 8. Benesty, J.; Chen, J.; Huang, Y.; Cohen, I. Pearson correlation coefficient. In *Noise Reduction in Speech Processing*; Springer: Berlin/Heidelberg, Germany, 2009; pp. 1–4.
- 9. Conlon, T.; Ruskin, H.J.; Crane, M. Cross-correlation dynamics in financial time series. *Physica A* 2009, 388, 705–714. [CrossRef]
- 10. Matos, J.; Gama, S.; Ruskin, H.; Sharkasi, A.; Crane, M. Correlation of worldwide markets' entropies. *Centro* **2006**. Available online: https://citeseerx.ist.psu.edu/messages/downloadsexceeded.html (accessed on 18 August 2022)
- 11. Goodell, J.W.; Goutte, S. Diversifying equity with cryptocurrencies during COVID-19. *Int. Rev. Financ. Anal.* **2021**, *76*, 101781. [CrossRef]
- 12. Mallinger-Dogan, M.; Szigety, M.C. Higher-Frequency Analysis of Low-Frequency Data. J. Portf. Manag. 2014, 41, 121–138. [CrossRef]
- 13. Brauneis, A.; Mestel, R. Price discovery of cryptocurrencies: Bitcoin and beyond. Econ. Lett. 2018, 165, 58-61. [CrossRef]
- 14. Caferra, R.; Vidal-Tomás, D. Who raised from the abyss? A comparison between cryptocurrency and stock market dynamics during the COVID-19 pandemic. *Financ. Res. Lett.* **2021**, *43*, 101954. [CrossRef]
- 15. Baur, D.G.; Dimpfl, T. The volatility of Bitcoin and its role as a medium of exchange and a store of value. *Empir. Econ.* **2021**, 61, 2663–2683. [CrossRef]
- 16. Barua, S. Understanding Coronanomics: The Economic Implications of the Coronavirus (COVID-19) Pandemic. 2020. Available online: https://papers.ssrn.com/sol3/papers.cfm?abstract\_id=3566477 (accessed on 17 July 2022).
- 17. Padhan, R.; Prabheesh, K. The economics of COVID-19 pandemic: A survey. Econ. Anal. Policy 2021, 70, 220–237. [CrossRef]
- Yousfi, M.; Zaied, Y.B.; Cheikh, N.B.; Lahouel, B.B.; Bouzgarrou, H. Effects of the COVID-19 pandemic on the US stock market and uncertainty: A comparative assessment between the first and second waves. *Technol. Forecast. Soc. Chang.* 2021, 167, 120710. [CrossRef]
- 19. Lahmiri, S.; Bekiros, S. The impact of COVID-19 pandemic upon stability and sequential irregularity of equity and cryptocurrency markets. *Chaos Solit. Fractals* **2020**, *138*, 109936. [CrossRef]
- 20. Kwapień, J.; Wątorek, M.; Drożdż, S. Cryptocurrency Market Consolidation in 2020 & 2021. Entropy 2021, 23, 1674. [CrossRef]
- 21. Drożdź, S.; Kwapień, J.; Oświęcimka, P.; Stanisz, T.; Wątorek, M. Complexity in economic and social systems: Cryptocurrency market at around COVID-19. *Entropy* 2020, 22, 1043. [CrossRef]
- 22. Nie, C.X. Analysis of critical events in the correlation dynamics of cryptocurrency market. *Phys. A Stat. Mech. Its Appl.* **2022**, 586, 126462. [CrossRef]
- 23. Miceli, M.A.; Susinno, G. Ultrametricity in fund of funds diversification. *Physica A* 2004, 344, 95–99. [CrossRef]
- 24. Onnela, J.P.; Chakraborti, A.; Kaski, K.; Kertesz, J. Dynamic asset trees and Black Monday. Physica A 2003, 324, 247–252. [CrossRef]
- Liu, H.; Zou, J.; Ravishanker, N. Clustering high-frequency financial time series based on information theory. *Appl. Stoch. Model. Bus. Ind.* 2022, *38*, 4–26. [CrossRef]
- 26. Durcheva, M.; Tsankov, P. Analysis of similarities between stock and cryptocurrency series by using graphs and spanning trees. In *AIP Conference Proceedings*; AIP Publishing LLC: Melville, NY, USA, 2019; Volume 2172, p. 090004.
- Katsiampa, P.; Yarovaya, L.; Zięba, D. High-Frequency connectedness between bitcoin and other top-traded crypto assets during the COVID-19 crisis. J. Int. Financ. Mark. Inst Money 2022, 79, 101578. [CrossRef]

- Francés, C.J.; Carles, P.; Arellano, D.J. The cryptocurrency market: A network analysis. *Esic Mark. Econ. Bus. J.* 2018, 49, 569–583.
   [CrossRef]
- 29. Briola, A.; Aste, T. Dependency structures in cryptocurrency market from high to low frequency. arXiv 2022, arXiv:2206.03386.
- 30. Gavin, J.; Crane, M. Community Detection in Cryptocurrencies with Potential Applications to Portfolio Diversification. *arXiv* **2021**, arXiv:2108.09756.
- 31. Yu, A.; Bünz, B. Community Detection and Analysis in the Bitcoin Network, CS 224W Final Report; Stanford University: Stanford, CA, USA, 2015.
- Atiya, H.R.; Nawaf, H.N. Prediction of Link Weight of bitcoin Network by Leveraging the Community Structure. In IOP Conference Series: Materials Science and Engineering; IOP Publishing: Bristiol, UK, 2020; Volume 928, p. 032080.
- 33. Jin, D.; Yu, Z.; Jiao, P.; Pan, S.; He, D.; Wu, J.; Yu, P.; Zhang, W. A survey of community detection approaches: From statistical modeling to deep learning. *IEEE Trans. Knowl. Data. Eng* **2021**. [CrossRef]
- 34. Cao, K.H.; Li, Q.; Liu, Y.; Woo, C.K. COVID-19's adverse effects on a stock market index. *Appl. Econ. Lett.* **2021**, *28*, 1157–1161. [CrossRef]
- Ali, M.; Alam, N.; Rizvi, S.A.R. Coronavirus (COVID-19)—An epidemic or pandemic for financial markets. J. Behav. Exp. Financ. 2020, 27, 100341. [CrossRef]
- 36. De Luca, G.; Loperfido, N. A skew-in-mean GARCH model. In *Skew-Elliptical Distributions and Their Applications: A Journey beyond Normality*; Genton, M.G., Ed.; Chapman and Hall/CRC: London, UK, 2004; pp. 233–252.
- 37. Karamti, C.; Belhassine, O. COVID-19 pandemic waves and global financial markets: Evidence from wavelet coherence analysis. *Financ. Res. Lett.* **2022**, *45*, 102136. [CrossRef] [PubMed]
- 38. Zhang, W.; Hamori, S. Crude oil market and stock markets during the COVID-19 pandemic: Evidence from the US, Japan, and Germany. *Int. Rev. Financ. Anal.* 2021, 74, 101702. [CrossRef]
- Adekoya, O.B.; Oliyide, J.A. How COVID-19 drives connectedness among commodity and financial markets: Evidence from TVP-VAR and causality-in-quantiles techniques. *Resour. Policy* 2021, 70, 101898. [CrossRef] [PubMed]
- 40. Chen, C.; Liu, L.; Zhao, N. Fear sentiment, uncertainty, and bitcoin price dynamics: The case of COVID-19. *Emerg. Mark. Financ. Trade* **2020**, *56*, 2298–2309. [CrossRef]
- 41. Baig, A.S.; Butt, H.A.; Haroon, O.; Rizvi, S.A.R. Deaths, panic, lockdowns and US equity markets: The case of COVID-19 pandemic. *Financ. Res. Lett* **2021**, *38*, 101701. [CrossRef]
- 42. Sapkota, N. News-based sentiment and bitcoin volatility. Int. Rev. Financ. Anal. 2022, 82, 102183. [CrossRef]
- 43. Bianchi, D. Cryptocurrencies as an asset class? An empirical assessment. J. Altern. Investments 2020, 23, 162–179. [CrossRef]
- 44. Anamika, A.; Subramaniam, S. Do news headlines matter in the cryptocurrency market? Appl. Econ. 2022, 1–17. [CrossRef]
- Vidal-Tomás, D. Transitions in the cryptocurrency market during the COVID-19 pandemic: A network analysis. *Financ. Res. Lett.* 2021, 43, 101981. [CrossRef]
- Assaf, A.; Bhandari, A.; Charif, H.; Demir, E. Multivariate long memory structure in the cryptocurrency market: The impact of COVID-19. Int. Rev. Financ. Anal. 2022, 82, 102132. [CrossRef]
- 47. Masconi, K.L.; Matsha, T.E.; Erasmus, R.T.; Kengne, A.P. Effects of different missing data imputation techniques on the performance of undiagnosed diabetes risk prediction models in a mixed-ancestry population of South Africa. *PLoS ONE* **2015**, *10*, e0139210. [CrossRef] [PubMed]
- Van der Heijden, G.J.; Donders, A.R.T.; Stijnen, T.; Moons, K.G. Imputation of missing values is superior to complete case analysis and the missing-indicator method in multivariable diagnostic research: A clinical example. *J. Clin. Epidemiol.* 2006, 59, 1102–1109. [CrossRef] [PubMed]
- 49. Genolini, C.; Jacqmin-Gadda, H. Copy mean: A new method to impute intermittent missing values in longitudinal studies. *Open J. Stat.* **2013**, *3*, 26. [CrossRef]
- 50. Rydberg, T.H. Realistic statistical modelling of financial data. Int. Stat. Rev. 2000, 68, 233–258. [CrossRef]
- 51. Amien, I.; Rajaratnam, K.; Kruger, R. Inference of aggregational gaussianity in asset returns exhibiting a paretian-distribution. *Procedia Econ. Financ.* **2015**, *25*, 400–407. [CrossRef]
- 52. Kratz, M.; Resnick, S.I. The QQ-estimator and heavy tails. Stoch. Model. 1996, 12, 699–724. [CrossRef]
- 53. Abdi, H.; Molin, P. Lilliefors/Van Soest's test of normality. In *Encyclopedia of Measurement and Statistics*; SAGE Publications: Thousand Oaks, OK, USA, 2007; pp. 540–544.
- 54. Takaishi, T. Time-varying properties of asymmetric volatility and multifractality in Bitcoin. *PLoS ONE* **2021**, *16*, e0246209. [CrossRef]
- 55. Kulyatin, I. Stylized Facts for Cryptocurrencies: A Sectoral Analysis; Medium: New York, NY, USA, 2015.
- 56. Tsay, R.S. Analysis of Financial Time Series; John Wiley & Sons: Hoboken, NJ, USA, 2005.
- 57. Schäfer, R.; Guhr, T. Local normalization: Uncovering correlations in non-stationary financial time series. *Physica A* **2010**, *389*, 3856–3865. [CrossRef]
- 58. Tjøstheim, D.; Otneim, H.; Støve, B. Statistical Dependence: Beyond Pearson's ρ. Stat. Sci. 2022, 37, 90–109. [CrossRef]
- 59. Wissler, C. The Spearman correlation formula. *Science* 1905, 22, 309–311. [CrossRef]
- 60. Lieberson, S. Limitations in the application of non-parametric coefficients of correlation. *Am. Sociol. Rev.* **1964**, *29*, 744–746. [CrossRef]

- 61. Plerou, V.; Gopikrishnan, P.; Rosenow, B.; Amaral, L.N.; Stanley, H.E. A random matrix theory approach to financial crosscorrelations. *Physica A* **2000**, *287*, 374–382. [CrossRef]
- 62. Cacciapuoti, C.; Maltsev, A.; Schlein, B. Local Marchenko-Pastur law at the hard edge of sample covariance matrices. *J. Math. Phys.* **2013**, *54*, 043302. [CrossRef]
- Laloux, L.; Cizeau, P.; Potters, M.; Bouchaud, J.P. Random matrix theory and financial correlations. *Int. J. Theor. Appl. Financ.* 2000, 3, 391–397. [CrossRef]
- 64. Mai, T.; Martin, C.; Marija, B. Student behaviours in using learning resources in higher education: How do behaviours reflect success in programming education? In Proceedings of the 7th International Conference on Higher Education Advances, Torino, Italy, 22–23 June 2021; pp. 47–55.
- 65. Plerou, V.; Gopikrishnan, P.; Rosenow, B.; Amaral, L.A.N.; Guhr, T.; Stanley, H.E. Random matrix approach to cross correlations in financial data. *Phys. Rev. E* 2002, *65*, 066126. [CrossRef]
- 66. Dimpfl, T.; Peter, F.J. Nothing but noise? Price discovery across cryptocurrency exchanges. *J. Financ. Mark.* **2021**, *54*, 100584. [CrossRef]
- 67. Sarr, A.; Lybek, T. Measuring Liquidity in Financial Markets. 2002. Available online: https://papers.ssrn.com/sol3/papers.cfm? abstract\_id=880932 (accessed on 27 July 2022).
- Li, T.; Shin, D.; Wang, B. Cryptocurrency Pump-and-Dump Schemes. 2021. Available online: https://papers.srn.com/sol3 /papers.cfm?abstract\_id=3267041 (accessed on 5 August 2022).
- 69. Auer, R.; Claessens, S. Regulating cryptocurrencies: Assessing market reactions. *BIS Q. Rev. Sept.* **2018**. Available online: https://papers.ssrn.com/sol3/papers.cfm?abstract\_id=3288097 (accessed on 8 August 2022).
- 70. Mai, T.T.; Bezbradica, M.; Crane, M. Learning behaviours data in programming education: Community analysis and outcome prediction with cleaned data. *Future Gener. Comput. Syst.* 2022, 127, 42–55. [CrossRef]
- 71. Zhang, S.; Lee, J.H. Analysis of the main consensus protocols of blockchain. ICT Express 2020, 6, 93–97. [CrossRef]
- 72. da Gama Silva, P.V.J.; Klotzle, M.C.; Pinto, A.C.F.; Gomes, L.L. Herding behavior and contagion in the cryptocurrency market. *J. Behav. Exp. Financ.* **2019**, 22, 41–50. [CrossRef]
- 73. Burda, Z.; Jarosz, A. Cleaning large-dimensional covariance matrices for correlated samples. *Phys. Rev. E* 2022, 105, 034136. [CrossRef] [PubMed]
- 74. Bouchaud, J.P.; Potters, M. Financial applications of random matrix theory: A short review. arXiv 2009, arXiv:0910.1205.
- Ledoit, O.; Wolf, M. Nonlinear shrinkage estimation of large-dimensional covariance matrices. Ann. Stat. 2012, 40, 1024–1060. [CrossRef]
- Bun, J.; Allez, R.; Bouchaud, J.P.; Potters, M. Rotational invariant estimator for general noisy matrices. *IEEE Trans. Inf. Theory* 2016, 62, 7475–7490. [CrossRef]
- 77. Ledoit, O.; Wolf, M. A well-conditioned estimator for large-dimensional covariance matrices. *J. Multivar. Anal.* **2004**, *88*, 365–411. [CrossRef]
- 78. Bun, J.; Knowles, A. An optimal rotational invariant estimator for general covariance matrices: The outliers. *Preprint* 2018. Available online: https://www.researchgate.net/profile/Joel-Bun/publication/323255675\_An\_Optimal\_Rotational\_Invariant\_ Estimator\_for\_General\_Covariance\_Matrices\_the\_outliers/links/5a89e7fba6fdcc6b1a424f88/An-Optimal-Rotational-Invariant-Estimator-for-General-Covariance-Matrices-the-outliers.pdf (accessed on 8 August 2022).
- Shcherbakov, M.V.; Brebels, A.; Shcherbakova, N.L.; Tyukov, A.P.; Janovsky, T.A.; Kamaev, V.A. A survey of forecast error measures. World Appl. Sci. J. 2013, 24, 171–176.
- 80. Markowitz, R. Cleaning Correlation Matrices. 2016. Available online: https://www.cfm.fr/assets/ResearchPapers/2016 -Cleaning-Correlation-Matrices.pdf (accessed on 8 August 2022).
- Conlon, T.; Ruskin, H.J.; Crane, M. Random matrix theory and fund of funds portfolio optimisation. *Physica A* 2007, 382, 565–576. [CrossRef]
- Yang, L.; McKay, M.R.; Couillet, R. High-dimensional MVDR beamforming: Optimized solutions based on spiked random matrix models. *IEEE Trans. Signal Process* 2018, 66, 1933–1947. [CrossRef]
- 83. Heimo, T.; Kaski, K.; Saramäki, J. Maximal spanning trees, asset graphs and random matrix denoising in the analysis of dynamics of financial networks. *Physica A* 2009, *388*, 145–156. [CrossRef]
- 84. de Prado, M.M.L. Machine Learning for Asset Managers; Cambridge University Press: Cambridge, UK, 2020.
- 85. O'Searcoid, M. Metric Spaces; Springer Science & Business Media: Berlin/Heidelberg, Germany, 2006.
- 86. Graham, R.L.; Hell, P. On the history of the minimum spanning tree problem. IEEE Ann. Hist. Comput. 1985, 7, 43–57. [CrossRef]
- Ghosh, B.; Motagh, M.; Haghighi, M.H.; Vassileva, M.S.; Walter, T.R.; Maghsudi, S. Automatic detection of volcanic unrest using blind source separation with a minimum spanning tree based stability analysis. *IEEE J. Sel. Top. Appl. Earth Obs. Remote Sens.* 2021, 14, 7771–7787. [CrossRef]
- 88. Denkowska, A.; Wanat, S. Linkages and systemic risk in the European insurance sector. New evidence based on Minimum Spanning Trees. *Risk Manag.* 2022, 24, 123–136. [CrossRef]
- 89. Coelho, R.; Gilmore, C.G.; Lucey, B.; Richmond, P.; Hutzler, S. The evolution of interdependence in world equity markets— Evidence from minimum spanning trees. *Physica A* **2007**, *376*, 455–466. [CrossRef]
- Kazemilari, M.; Mohamadi, A.; Mardani, A.; Streimikis, J. Network topology of renewable energy companies: Minimal spanning tree and sub-dominant ultrametric for the American stock. *Technol. Econ. Dev. Econ.* 2019, 25, 168–187. [CrossRef]

- 91. Nguyen, Q.; Nguyen, N.; Nguyen, L. Dynamic topology and allometric scaling behavior on the Vietnamese stock market. *Physica* A **2019**, *514*, 235–243. [CrossRef]
- Huang, C.; Zhao, X.; Su, R.; Yang, X.; Yang, X. Dynamic network topology and market performance: A case of the Chinese stock market. Int. J. Financ. Econ. 2022, 27, 1962–1978. [CrossRef]
- 93. Giudici, P.; Pagnottoni, P.; Polinesi, G. Network models to enhance automated cryptocurrency portfolio management. *Front. Artif. Intell.* **2020**, *3*, 22. [CrossRef]
- 94. Kruskal, J.B. On the shortest spanning subtree of a graph and the traveling salesman problem. *Proc. Am. Math. Soc.* **1956**, *7*, 48–50. [CrossRef]
- Huang, F.; Gao, P.; Wang, Y. Comparison of Prim and Kruskal on Shanghai and Shenzhen 300 Index hierarchical structure tree. In Proceedings of the 2009 International Conference on Web Information Systems and Mining, Shanghai, China, 7–8 November 2009; pp. 237–241.
- 96. Mantegna, R.N. Hierarchical structure in financial markets. Eur. Phys. J. B 1999, 11, 193–197. [CrossRef]
- 97. Zhao, B.; Yang, W.; Wen, J.; Zhang, W. The Financial Market in China under the COVID-19. *Emerg. Mark. Financ. Trade* 2022, 1–13. [CrossRef]
- 98. Brida, J.G.; Risso, W.A. Hierarchical structure of the German stock market. *Expert Syst. Appl.* 2010, 37, 3846–3852. [CrossRef]
- 99. MacMahon, M.; Garlaschelli, D. Community detection for correlation matrices. *arXiv* 2013, arXiv:1311.1924.
- Chakrabarti, P.; Jawed, M.S.; Sarkhel, M. COVID-19 pandemic and global financial market interlinkages: A dynamic temporal network analysis. *Appl. Econ.* 2021, *53*, 2930–2945. [CrossRef]
- Bazzi, M.; Porter, M.A.; Williams, S.; McDonald, M.; Fenn, D.J.; Howison, S.D. Community detection in temporal multilayer networks, with an application to correlation networks. *Multiscale Model. Simul.* 2016, 14, 1–41. [CrossRef]
- 102. Anghinoni, L.; Vega-Oliveros, D.A.; Silva, T.C.; Zhao, L. Time series pattern identification by hierarchical community detection. *Eur. Phys. J. Spec. Top.* **2021**, 230, 2775–2782. [CrossRef]
- 103. Raghavan, U.N.; Albert, R.; Kumara, S. Near linear time algorithm to detect community structures in large-scale networks. *Phys. Rev. E* 2007, *76*, 036106. [CrossRef]
- 104. Pujol, J.M.; Erramilli, V.; Rodriguez, P. Divide and conquer: Partitioning online social networks. arXiv 2009, arXiv:0905.4918.
- Roma, G.; Herrera, P. Community structure in audio clip sharing. In Proceedings of the 2010 INCoS, Thessaloniki, Greece, 24–26 November 2010; pp. 200–205.
- 106. Wang, C.; Wang, F. GIS-automated delineation of hospital service areas in Florida: From Dartmouth method to network community detection methods. *Ann. GIS* 2022, *28*, 93–109. [CrossRef]
- 107. Raeder, T.; Chawla, N.V. Market basket analysis with networks. Soc. Netw. Anal. Min. 2011, 1, 97–113. [CrossRef]
- 108. Blondel, V.D.; Guillaume, J.L.; Lambiotte, R.; Lefebvre, E. Fast unfolding of communities in large networks. *J. Stat. Mech. Theory Exp.* **2008**, 2008, P10008. [CrossRef]
- 109. Newman, M.E.; Girvan, M. Finding and evaluating community structure in networks. *Phys. Rev. E* 2004, 69, 026113. [CrossRef] [PubMed]
- 110. Boyle, M.; Bellucco-Chatham, A. 2008 Recession: What the Great Recession Was and What Caused It; Investopedia: New York, NY, USA, 2022.
- Liang, J.; Li, L.; Zeng, D.; Zhao, Y. Correlation-based dynamics and systemic risk measures in the cryptocurrency market. In Proceedings of the IEEE 2018 ISI, Miami, FL, USA, 9–11 November 2018; pp. 43–48.
- 112. Spelta, A.; Araújo, T. The topology of cross-border exposures: Beyond the minimal spanning tree approach. *Physica A* **2012**, 391, 5572–5583. [CrossRef]
- 113. Feng, W.; Wang, Y.; Zhang, Z. Can cryptocurrencies be a safe haven: A tail risk perspective analysis. *Appl. Econ.* **2018**, 50, 4745–4762. [CrossRef]
- 114. Corbet, S.; Lucey, B.; Urquhart, A.; Yarovaya, L. Cryptocurrencies as a financial asset: A systematic analysis. *Int. Rev. Financ. Anal* **2019**, *62*, 182–199. [CrossRef]
- 115. Ammous, S. Can cryptocurrencies fulfil the functions of money? Q. Rev. Econ. Financ. 2018, 70, 38-51. [CrossRef]
- 116. Klein, T.; Thu, H.P.; Walther, T. Bitcoin is not the New Gold–A comparison of volatility, correlation, and portfolio performance. *Int. Rev. Financ. Anal.* **2018**, *59*, 105–116. [CrossRef]
- 117. Su, F.; Wang, X.; Yuan, Y. The intraday dynamics and intraday price discovery of bitcoin. *Res. Int. Bus. Financ.* **2022**, *60*, 101625. [CrossRef]
- 118. Megaritis, A.; Vlastakis, N.; Triantafyllou, A. Stock market volatility and jumps in times of uncertainty. *J. Int. Money Financ.* 2021, 113, 102355. [CrossRef]
- 119. Aslam, F.; Ferreira, P.; Mughal, K.S.; Bashir, B. Intraday volatility spillovers among European financial markets during COVID-19. *Int. J. Financ. Stud.* **2021**, *9*, 5. [CrossRef]
- 120. Jarosław, K.; Marcin, W.; Marija, B.; Martin, C.; Mai, T.; Stanisław, D. Analysis of inter-transaction time fluctuations in the cryptocurrency market. *Chaos Interdiscip. J. Nonlinear Sci.* 2022, 32, 083142.
- Rosenberg, A.; Hirschberg, J. V-measure: A conditional entropy-based external cluster evaluation measure. In Proceedings of the 2007 EMNLP-CoNLL, Prague, Czech Republic, 28–30 June 2007; pp. 410–420.
- 122. Liang, J.; Li, L.; Zeng, D. Evolutionary dynamics of cryptocurrency transaction networks: An empirical study. *PLoS ONE* **2018**, 13, e0202202.

- 123. Koutra, D.; Parikh, A.; Ramdas, A.; Xiang, J. Algorithms for Graph Similarity and Subgraph Matching. 2011. Volume 17. Available online: https://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.377.4291&rep=rep1&type=pdf (accessed on 8 August 2022).
- 124. Gera, R.; Alonso, L.; Crawford, B.; House, J.; Mendez-Bermudez, J.; Knuth, T.; Miller, R. Identifying network structure similarity using spectral graph theory. *Appl. Netw. Sci.* 2018, *3*, 2. [CrossRef]
- 125. Yarovaya, L.; Matkovskyy, R.; Jalan, A. The effects of a'Black Swan'event (COVID-19) on herding behavior in cryptocurrency markets: Evidence from cryptocurrency USD, EUR, JPY and KRW Markets. J. Int. Financ. Mark. Inst. Money 2021, 75, 101321. [CrossRef]
- 126. Youssef, M.; Waked, S.S. Herding behavior in the cryptocurrency market during COVID-19 pandemic: The role of media coverage. N. Am. J. Econ. Financ. 2022, 62, 101752. [CrossRef]
- 127. Susana, D.; Kavisanmathi, J.; Sreejith, S. Does herding behaviour among traders increase during COVID 19 pandemic? Evidence from the cryptocurrency market. In Proceedings of the International Working Conference on Transfer and Diffusion of IT, Tiruchirappalli, India, 18–19 December 2020; Springer: Berlin/Heidelberg, Germany, 2020; pp. 178–189.
- 128. Kaiser, L.; Stöckl, S. Cryptocurrencies: Herding and the transfer currency. Financ. Res. Lett 2020, 33, 101214. [CrossRef]
- 129. Mnif, E.; Salhi, B.; Mouakha, K.; Jarboui, A. Investor behavior and cryptocurrency market bubbles during the COVID-19 pandemic. *Rev. Behav. Financ.* 2022, *ahead-of-print.* [CrossRef]
- Corbet, S.; Larkin, C.; Lucey, B.; Yarovaya, L. KODAKCoin: A blockchain revolution or exploiting a potential cryptocurrency bubble? *Appl. Econ. Lett* 2020, 27, 518–524. [CrossRef]
- 131. Huber, C.; Huber, J.; Kirchler, M. Market shocks and professionals' investment behavior—Evidence from the COVID-19 crash. *J. Bank. Financ.* **2021**, *133*, 106247. [CrossRef]